Author's response to reviews

Title: Analysis of neonatal mortality: is standardizing for relative birth weight biased?

Authors:

Dr Robert W Platt (robert.platt@mcgill.ca)
Prof Cande V Ananth (cande.ananth@umdnj.edu)
Michael S Kramer (michael.kramer@mcgill.ca)

Version: 4 Date: 4 Jun 2004

I hope I haven't missed any further editorial corrections!